Author's response to reviews

Title: All-Arthroscopic versus Mini-Open Repair of Small or Moderate Sized Rotator Cuff Tears: A Protocol for a Randomized Trial [NCT00128076] - A study protocol

Authors:

Joy C MacDermid (macderj@mcmaster.ca)
Richard Holtby (richard.holtby@sw.ca)
Helen Razmjou (helen.razmjou@sw.ca)
Dianne Bryant (jackowdm@mcmaster.ca)
Joints Canada (jointscanada@jointscanada.ca)

Version: 2 Date: 2 February 2006

Author's response to reviews:

Ethics and funding approvals received